Author's response to reviews

Title: Validation of the 10-item Chinese Perceived Stress Scale in Elderly Service Workers: One-factor versus Two-factor Structure

Authors:

Siu-man Ng (ngsiuman@hku.hk)

Version: 2 Date: 27 September 2012

Author's response to reviews:

As requested, the following statements were added to the Methods section. "Voluntary participation was ensured throughout the study and written informed consent was collected from each participant. Ethical approval for the study was granted by the Human Research Ethics Committee for Non-Clinical Facilities of the The University of Hong Kong (reference number EA171210)."